MEG Management Committee meeting

18.02.2009, 14:00hrs, West Wing Seminar Room, CBU

Minutes
Present:  Friedemann Pulvermuller, Gary Chandler, Clare Dine, Rik Henson, Oleg Korzyukov, Olaf Hauk, Maarten Van Casteren, Marie Smith, Rob Anthony, Georgie Morrill.

1. Apologies for absences
· Yury Shtyrov
2. Matters arising 
OK brought up the issue with the door in the corridor outside the MEG lab.  It actually causes artefacts in the recordings because it is regularly used and swings back quite fast which OK thinks may cause some vibrations through the walls. FP wondered if there might be magnets in the door causing this too so OK is going to liaise with GC and sort out.  It has a fire door hinge at the top so it maybe as easy as simply adjusting this to slow it down when closing.
Action: OK to liaise with GC about door.
3. Minutes of the last meeting
Gold Study: FP questioned OK further about the running time of this experiment and OK informed the committee that it would only take about 35mins. 

The minutes were approved pending some minor amendments.
Action: FP, OK and OH meet to discuss further details. 

4. Computing








RA, and RH reported that they still haven’t heard about when the new version of Maxfilter is due out.  FP asked that RA looks into this. 

YS sent an email prior to the meeting (as he was absent at this) asking about HP the booting problems with the user room computers.  RA will look into this and check the progress of moving the HP computer from the MEG lab and putting a normal desktop in with a general MEG account.
File space: Computing have moved the heavy MEG users across which means you can have more room for your data on the disc space without moving others’ data off. RH asked RA to look into the average life span of the files because of the amount of ‘churn’.  RA informed the committee that there could be quite a lot of churn with have approximately 12TB of space on there. FP is still using the normal imaging disk space so RA advised it would be good to move the MEG data over to the new disk space which has more room.  RC also advised that there is only about 2.2 TB on the ERP drive so isn’t wise to use this either as will push other users’ data off.  OH asked RA about the little bit of disc space for testing data for MEG for empty room/phantom recordings, and reported that he can’t seem to connect to this directory. Apparently this is solved through remapping and OH is going to have another go.  RC informed RA that he has problems going from his imaging, mounting onto someone else’s and then going back to his own as it doesn’t log out of theirs.  RA replied that Windows has issues with connecting to the same server resource but with lots of different accounts. RA suggested to OH that the best thing to do would be to create a separate group area and put Samba mapping which would route it faster.  This could be called ‘MEG Data’.
Action: RA to look into booting of HP workstations, find out about Max filter software and sort out normal desktop PC in MEG lab.
5. Lab business and progress
5.1 HPI and ‘Kaptah 1’ failure
CD reported that the Kaptah 1 failure was happening about twice per week, and MvC has contacted Neuromag about it.  RH pointed out that this happened last year and seemed to be fixed okay.  FP suggested that it could be a data transfer problem and he will speak to YS about this.  With the HPI problems, GC reported that Neuromag changed the cable which the coils plug into and if there are further problems they will change the plug as it seems to be lower quality compared to the rest. 

5.2 Wall for sound insulation

GC informed the committee that the drawings were being changed and he was just waiting to receive the new ones.  
5.3 Empty room/phantom recordings for quality control
OH made a suggestion to have routine checkups to ensure the machine is running as it should.  This used to be done in the past but there was no one free to analyse it at the time.  The committee agreed that empty room and phantom recordings would cover all aspects of quality control with the machine.  OH would go through this with the operators so they can carry this out in the morning before each day of recording.
5.4 Other issues
OK brought up the issue that he needs new electrodes for separate EEG work and has ordered 4 new ones via CD.
Action: MvC and GC to look at HPI/Kaptah 1 failure to make sure there are no issues. OH go through empty room/phantom recordings with operators.
6. MEG lab stats for January/February
GM showed the committee the lab stats and informed them that the MEG lab was filling up again gradually.  RH brought up the fact that one of the researchers from The Faculty of Education has only paid for a certain amount of slots so we need to keep an eye on this.  It maybe that they didn’t use all of their slots from last year so some of these can be carried over.  
Action: GM to check how many slots have been used for Faculty of Ed and bring a spreadsheet along to the next meeting showing how many slots have been paid for by the partners.  GM to produce bullet point summary of lab for the IMC/UMC.
7. AOB: 
None.
9. Date of the next meeting:

Wednesday 18thth March at 2:00pm in the WWSR.
